Author’s response to reviews

Title: An experimental animal model of aseptic loosening of hip prostheses in sheep to study early biochemical changes at the interface membrane

Authors:

  Alexander El Warrak (warrakel@uiuc.edu)
  Marvin Olmstead (thrdvm@worldnet.att.net)
  Rebecca Schneider-Hafliger (rebecca.s@bluewin.ch)
  Lorenz Meinel (lorenz.meinel@pharma.ethz.ch)
  Regula Bettschart-Wolfisberger (rbettschart@vetclinics.unizh.ch)
  Dr Margret K Akens (makens@ovc.uoguelph.ca)
  Jorg A Auer (jauer@vetclinics.unizh.ch)
  Brigitte von Rechenberg (bvrechenberg@vetclinics.unizh.ch)

Version: 3 Date: 28 Feb 2004

Line numbers have been removed. The author’s contribution has been placed behind the acknowledgement section.